Author's response to reviews

Title: Rare recessive loss-of-function methionyl-tRNA synthetase mutations presenting as a multi-organ phenotype

Authors:

Eline v Meel (hvanmeel@dom.wustl.edu)
Daniel J Wegner (wegner_d@kids.wustl.edu)
Paul Cliften (pcliften@genetics.wustl.edu)
Marcia C Willing (willing_m@kids.wustl.edu)
Frances White (fwhite@path.wustl.edu)
Stuart Kornfeld (skornfel@dom.wustl.edu)
F. Sessions Cole (cole@kids.wustl.edu)

Version: 3 Date: 11 September 2013

Author's response to reviews: see over